Author’s response to reviews

Title: Identification and characterization of a novel 43-bp deletion mutation of the ATP7B gene in a Chinese patient with Wilson's disease: a case report

Authors:
Zhengfeng Xu (zhengfeng_xu_nj@163.com)
Gang Liu (liugang_516@126.com)
Dingyuan Ma (dingyuanma2002@163.com)
Jian Cheng (13770571912@163.com)
Jingjing Zhang (18913384682@163.com)
Chunyu Luo (lcy0929@126.com)
Yun Sun (sunyun80@126.com)
Ping Hu (njfybjyhuping@163.com)
Yuguo Wang (wyg.num1@163.com)
Tao Jiang (jiangzhang784@163.com)

Version: 3 Date: 19 Mar 2018

Author’s response to reviews:

Response to Editor’ Comments

Editor Comment 1: Your Consent for publication statement still does not state whether the consent was written or verbal ("The parent of the patient consented to the publication of medical data and genetic data.") This information is mandatory according to our editorial policies, especially for Case Reports which are identifying by their very nature.

Response: It was added that “The parent of the patient provided written informed consent for publication of medical data and genetic data”. (line 250-251, Page 12)